Author's response to reviews

Title: Binocular function to increase visual outcome in patients implanted with a diffractive trifocal intraocular lens

Authors:

Florian TA Kretz (mail@florian-kretz.de)
Matthias Müller (m.mueller@augenklinik.de)
Matthias Gerl (m.gerl@augenklinik.de)
Ralf H Gerl (r.gerl@augenklinik.de)
Gerd U Auffarth (gerd.auffarth@med.uni-heidelberg.de)

Version: 2
Date: 22 February 2015

Author's response to reviews:

Dear Editor, we have performed the following changes regarding your recommendations:

1. In the methods section: page 6, line 117-120: the approving ethics committee and the registration number have been added: “The ethics committee of the doctors chamber Westfalen-Lippe and the medical faculty of the Westfälischen Wilhelms-University, Münster approved the study under the reference number: 2014-078-b-S.”

2. The study was registered with the German Clinical Trials Register (DRKS) und following number: DRKS00007837